Author's response to reviews

Title: A Systematic Review of the Psychometric Properties of Self-Report Research Utilization Measures used in Healthcare

Authors:

Janet E Squires (jasquires@ohri.ca)
Carole A Estabrooks (carole.estabrooks@ualberta.ca)
Hannah M O'Rourke (hierke@ualberta.ca)
Petter Gustavsson (petter.gustavsson@ki.se)
Christine V Newburn-Cook (christine.newburn-cook@ualberta.ca)
Lars Wallin (lars.wallin@karolinska.se)

Version: 4 Date: 1 May 2011

Author's response to reviews: see over
30 April 2011

Dr Michel Wensing
Associate Editor
Implementation Science

RE: MS: 7887793454385918
A Systematic Review of the Psychometric Properties of Instruments used to Measure Research Utilization in Healthcare

Dear Dr Wensing:

Thank-you for the opportunity to revise and resubmit the above-identified manuscript.

With respect to the following editorial comment: “When reading the paper I assumed all the time that the instruments are in fact questionnaires or self-report measures, but I missed an explicit statement saying this. If it is correct, I would like this to be included in the paper, abstract and perhaps also in the title.” Yes, we are referring to self-report measures. We have made this explicit in the title, abstract, and throughout the manuscript text as suggested. We have maintained track changes in the files for easy identification of these changes.

We have also changed the title of the manuscript to reflect this comment. The new title is: A Systematic Review of the Psychometric Properties of Self-Report Research Utilization Measures used in Healthcare

If you require further clarification please do not hesitate to contact me.

Sincerely,

Janet E Squires RN, PhD
Clinical Epidemiology Program
Ottawa Hospital Research Institute
Ottawa, Ontario, Canada

cc
Carole A. Estabrooks
Hannah M. O’Rourke
Petter Gustavsson
Christine V. Newburn-Cook
Lars Wallin